Author's response to reviews

Title: The ethics of talking about ‘HIV cure’

Authors:

Stuart Rennie (stuart_rennie@med.unc.edu)
Mark Siedner (MSEIDNER@mgh.harvard.edu)
Joseph D Tucker (jdtucker@med.unc.edu)
Keymanthri Moodley (km@sun.ac.za)

Version: 4 Date: 4 March 2015

Author's response to reviews:

Dear Editors,

Thank you for the positive response in regard to the publication of our manuscript, the “Ethics of Talking about ‘HIV cure’”, in BMC Medical Ethics.

In regard to the comments from the reviewers, both reviewers stated that we had adequately addressed their previous comments and concerns.

Per your email correspondence, we have added the email addresses of the authors on the title page.

Thank you again for your patience and diligence in regard to our manuscript.

Regards, Stuart Rennie

Stuart Rennie, PhD, MA
Department of Social Medicine,
UNC Bioethics Center,
University of North Carolina at Chapel Hill